Author’s response to reviews

Title: Gastroesophageal reflux and antacid therapy in IPF: Analysis from the Australia IPF Registry

Authors:
Helen Jo (helen.jo@sydney.edu.au)
Tamera Corte (tameracorte@mac.com)
Ian Glaspole (i.glaspole@alfred.org.au)
Christopher Grainge (Christopher.grainge@hnehealth.nsw.gov.au)
Peter Hopkins (peter.hopkins@health.qld.gov.au)
Yuben Moodley (yuben.moodley@uwa.edu.au)
Paul Reynolds (paul.reynolds@adelaide.edu.au)
Sally Chapman (sally.chapman@health.sa.gov.au)
E Haydn Walters (haydn.walters@utas.edu.au)
Christopher Zappala (christopher_Zappala@health.qld.gov.au)
Heather Allan (heather7475@gmail.com)
Gregory Keir (gregory.keir@health.qld.gov.au)
Wendy Cooper (wendy.cooper@health.nsw.gov.au)
Annabelle Mahar (annabelle.mahar@health.nsw.gov.au)
Samantha Ellis (s.ellis@alfred.org.au)
Sacha Macansh (sacha@lungfoundation.com.au)
Nicole Goh (nicole.goh@austin.org.au)

Version: 2 Date: 03 Apr 2019

Author’s response to reviews:

Dear Matthew Hickey,
Sorry for the slight delay. I have been away at a conference.

Please find the amended manuscript attached.

Helen